Author's response to reviews

Title: Genetic variability in the precore and core promoter regions of hepatitis B virus strains in Karachi

Authors:

Zaigham Abbas (zaigham@akunet.org)
Rana Muzaffar (rmuzafar@super.net.pk)
Arif Siddiqui (info@siut.org)
Syed AA Naqvi (siutpk@sat.net.pk)
Syed AH Rizvi (siutpk@sat.net.pk)

Version: 5 Date: 16 July 2006

Author's response to reviews:

Abstract- Background/aims changed to "Background".

Introduction - Renamed this section "Background".

References- The reference list now contains all authors names.